Author's response to reviews

Title: Managing the risk of cancer in Cowden Syndrome: a case report

Authors:

HAMMAMI Sonia (soniawhammami@yahoo.fr)
BERRICHE Olfa (olfaberriche@yahoo.fr)
BELHADJALI Hichem (Belhadjalihichem@yahoo.fr)
HELLARA Olfa (olfahellara@rns.tn)
FAROOQ Ansar (afarooq@nhs.net)
MAHJOUB Sylvia (sylviamahjoub@yahoo.fr)

Version: 11 Date: 3 April 2012

Author's response to reviews: see over
Letter to the Editor In Chief

Journal of Medical Case reports

Manuscript CE : 2071914211624960

Title: The key to managing the risk of cancer in Cowden Syndrome: a case report
Authors: Sonia Hammami, Olfa Berriche, Hichem Belhadj Ali, Olfa Hellara, A Farooq, Silvia Mahjoub

According to the editorial team suggestion, the authors would like to point out by the present document the changes reported in the third revised the above paper: (see correction and addition in blue characters)

1/ The titre was changed to be more descriptive: Managing the risk of cancer in Cowden Syndrome: a case report.

2/ Figure 1 was removed

3/ The paper was corrected by English medical editing.

With best regards,

Your sincerely